Title: A cluster randomized trial to assess the impact of opinion leader endorsed evidence summaries on improving quality of prescribing for patients with chronic cardiovascular disease: rationale and design [ISRCTN26365328]

Authors:

Sumit R. Majumdar (me2.majumdar@ualberta.ca)
Finlay A. McAlister (finlay.mcalister@ualberta.ca)
Ross T. Tsuyuki (ross.tsuyuki@ualberta.ca)

Version: 2 Date: 24 June 2005

Author's response to reviews:

All of the requested revisions were formatting changes only - all requested formatting changes have been completed to specification. In addition, the ISRCTN# has now been included in the manuscript.